The Ixodes scapularis genome paper published in Nature Communications has been cited at lines 116, 198, 203, 552, and 976 in the second revised manuscript. In addition, the GenBank project accession number has been added at line 552.

The accession number for our PRIDE dataset remains the same (PXD004997) without the DOI.It has been described in the Methods (line 671) and Data availability (line 676) sections.

DOI information for our dataset in GigaDB has been added as reference 100.